Recent Trends in Power Quality Improvement: Review of the Unified Power Quality Conditioner

Swati Gade^{1†}, Rahul Agrawal¹, and Ravindra Munje², Non-members

ABSTRACT

The electrical power sector is currently focusing on power quality (PQ) issues and their mitigation. Industrial automation and the use of power electronic converters for the integration of distributed generation create various PQ problems. This necessitates PQ enhancement, which in turn helps to improve the life span of the equipment as well as the reliability of supply for feeding critical loads within the system. This paper presents a comprehensive review of the Unified Power Quality Conditioner (UPQC) and its widespread application in the distribution system. The UPQC belongs to the family of active power filters. It contributes to the alleviation of voltage and current-related PQ issues along with power factor correction and the integration of renewable energy systems in the distribution network. This paper discusses various topologies, compensation methods, control theories, and the technological developments in recent years. More than 160 research papers have summarized the features of UPQC for further applications. Based on the outcomes of the investigation, the future direction of the UPQC is discussed. This paper is expected to play a major role in guiding research scholars in the application of the UPQC.

Keywords: Harmonics, Power Quality, Renewable Energy System, Solar PV, Unified Power Quality Conditioner

LIST OF SYMBOLS

1P2W	Single-Phase Two-Wire
2C	Split Capacitor
3-HB	Three H-Bridge
3P3W	Three-Phase Three-Wire

Manuscript received on May 2, 2021; revised on June 1, 2021; accepted on June 8, 2021. This paper was recommended by Associate Editor Kaan Kerdchuen.

¹The authors are with the Department of Electrical Engineering, Sandip University, Nashik, India.

²The author is with the Department of Electrical Engineering, K. K. Wagh Institute of Engineering Education and Research, Nashik, India.

 $^{\dagger}\mathrm{Corresponding}$ author: swatiagade@gmail.com

©2021 Author(s). This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivs 4.0 License. To view a copy of this license visit: https://creativecommons. org/licenses/by-nc-nd/4.0/.

Digital Object Identifier 10.37936/ecti-eec.2021193.244936

3P4W	Three-Phase Four-Wire
3L	Three Leg
3L-2C	Three-Leg Split Capacitor
$4\mathrm{L}$	Four Leg
ALO	Ant Lion Optimization
ANFIS	Adaptive Neuro-Fuzzy Inference
	System
ANN	Artificial Neural Network
APF	Active Power Filters
COA	Cuckoo Optimization Algorithm
CSC	Current Source Converter
CSOGI	Cascaded Second-Order General-
00001	ized Integrator
DA	Dragonfly Algorithm
DANE	Digital Adaptive Notch Filter
DF	Differential Evolution
DC	Distributed Concretion
DG	Distributed Generation
DSU	Delayed Signal Cancentation
DVD	Distribution Static Compensator
	Eineffer Algorithm
FA FLO	Fireny Algorithm
FLC	Fuzzy Logic Controller
GA	Genetic Algorithm
GOA	Grasshopper Optimization Algo-
CINO.	rithm
GWO	Grey Wolf Optimizer
LPF	Low Pass Filter
LS	Loss Sensitive
MOPSO	Multi-Objective Particle Swarm
	Optimization
MOWOA	Multi-Objective Whale Optimiza-
	tion Algorithm
MPPT	Maximum Power Point Tracking
PAC	Phase Angle Control
PCC	Point of Common Coupling
PEC	Power Electronic Converter
PI 	Proportional Integral
PID	Proportional Integral Derivative
PLL	Phase-Locked Loop
PQ	Power Quality
PSO	Particle Swarm Optimization
RES	Renewable Energy Sources
ROA	Rider Optimization Algorithm
SCA	Sine Cosine Algorithm
SFCL	Superconducting Fault Current
	Limiter
SOGI	Second-Order Generalized Integra-
	tor

Solar PV	Solar Photovoltaic
SRF	Synchronous Reference Frame
THD	Total Harmonic Distortion
UPFC	Unified Power Flow Controller
UPOC	Unified Power Quality Conditioner
UPOC-D	Distributed UPQC
UPOC-DG	Distributed Generation UPOC
UPOC-I	Interline UPOC
UPOC I	Loft Shunt UPOC
UPOC MC	Multi Convertor UPOC
UDOC MD	Madulan UDOC
UPQC-MD	Modular UPQC
UPQC-ML	Multilevel UPQC
UPQC-R	Right Shunt UPQC
UPQC-P	UPQC Active Power Control
UPQC-Q	UPQC Reactive Power Control
UPQC-S	UPQC Simultaneous Active and
	Reactive Power Control
UPQC-VA _{min}	UPQC Minimum Volt-Ampere (VA)
•	Loading
I cr m	Inductor Ripple Current
I_{π}	Inductor Current Ripple
I	Source Current During Normal
18	Conditions
т	Conditions
I_{s1}	Source Current During Sag/Swei
т	Conditions
I_{DSTAT}	Per Phase Current Injected by the
	DSTATCOM
k	Ratio Between Actual Source and
	Rated Source Voltages
P_{se}	Active Power by the DVR
P_{sh}	Active Power by the DSTATCOM
Q_{se}	Reactive Power Handled by the
	DVR
Q_{sh}	Reactive Power by the DSTAT-
0.010	COM
S	VA Loading of the DVB
S_{se}	VA Loading of the DSTSTACOM
Supera	Total VA Loading of the UPOC
SUPQC V	Lowest Value of the DC Link Valt
V_{dc1}	Lowest value of the DC-Link volt-
T 7	age
V_{DVR}	Voltage Injected by the DVR Dur-
	ing Sag/Swell Conditions
V_L	Rated Load Voltage
I_L	Rated Load Current
V_{ph}	Per Phase Voltage
V_s	Rated Source Voltage
I_s	Rated Source Current
δ	Power Angle
ϕ	Rated Load Power Factor Angle
γ	Displacement Angle Between the
1	Source and Series-Injected Volt-
	and perico-injected volt-
	ageo

1. INTRODUCTION

In the present days, the use of solid-state devices produces power quality (PQ) hitches, such as voltage sag, swell, power surges, notches, spikes, flicker, voltage unbalance, and harmonics in the supply

Table 1: Standards and PQ phenomena.

PQ Phenomena	Standard	Ref.
Classification of	IEC 61000-2-1: 1990	[5]
power quality	IEC 61000-2-5: 1995	[6]
	IEEE 1159: 1995	[7]
	IEC 816: 1984	[8]
Transients	IEC 61000-2-1: 1990	[5]
	IEEE c62.41: (1991)	[9]
	IEEE 1159: 1995	[7]
Voltage sag/swell	IEC 61000-2-1: 1990	[5]
and interruption	IEEE 1159: 1995	[7]
	IEC 61000-2-1: 1990	[5]
Harmonics	IEC 61000-4-7: 1991	[10]
	IEEE 519: 1992	[11]
Voltage flicker	IEC 61000-4-15: 1997	[12]

across the load end of the consumers [1–4]. The main objective of any electric power supply system is to provide a continuous sinusoidal voltage of constant magnitude and frequency with balanced sinusoidal currents. On the other hand, all significant and sensitive loads require uninterrupted sinusoidal, balanced voltage of constant magnitude, and frequency. Otherwise, the protection system malfunctions, causing substantial loss of data, time, product quality, and service [1]. To achieve power quality standardization, the International Electrotechnical Commission (IEC) and the Institute of Electrical and Electronics Engineers (IEEE) have framed different power quality standards. Table 1 lists these standards and their respective PQ phenomena.

The integration of distributed generation (DG) involving windmills, solar plants, fuel cells, etc., is growing considerably. The application of DG has numerous technical, environmental, and economical advantages. However, the integration of DG imposes many challenges on the electric utilities [13]. The controlling of DG mainly depends upon reactive power compensation. Frequent changes to the reactive power consumption of bulk loads can introduce voltage sag and swell into the system. This causes a change in the real power demand of the system, resulting in power fluctuations [14]. Uncompensated reactive power also affects the efficiency of DG system, power factor (PF), and active power capability. The use of a power electronic converter (PEC) for the interconnection of DG to the utility grid ensures the safe functioning of equipment and switching between However, it introduces a wide range of sources. PQ problems such as current and voltage harmonics, voltage sag/swell, voltage and current unbalance, voltage flicker, load reactive power, neutral current, impulse transients, and interruptions [15].

Active power filter (APF) technology is used for compensating voltage-current harmonics, reactive power, reducing flicker, and improving voltage balance in 3-phase AC systems [4, 16–17]. Various



Fig. 1: Literature reviewed for this article.

custom power devices such as the dynamic voltage restorer (DVR), distribution static compensator (DSTATCOM), and unified power quality conditioner (UPQC) which belong to the APF family are used to mitigate various PQ problems [3]. This study focuses on the UPQC, which is a combination of both DVR and DSTATCOM. It is considered to be an optimal device for critical and sensitive loads to mitigate both voltage and current-based power quality problems [4, 18–19]. The UPQC consists of two PECs linked to a common DC-link back to back. It is used in the distribution system for simultaneous series and shunt compensation, and should not be confused with the unified power flow controller (UPFC) employed in the transmission system. To balance the shunt/series compensation, UPFC works within a balanced and distortion-free transmission system environment. On the other hand, UPQC must work in an environment with unbalances and distortions created by DC components, voltage harmonics, and current harmonics [20]. The main objective of the UPQC is to reduce the injection of active power through both series and shunt APFs to minimize the circulating active power. Earlier published surveys on active power filters [16], the UPQC for power quality enhancement [20], and the UPQC role in distributed generation [14] provide excellent guidelines for researchers. The researchers in [16] present a survey on the classification of an APF based on its kVA rating and speed of response, circuit configuration, system parameter compensation, and control techniques. Khadkikar [20] provides a review on various topologies, configurations, methods of compensation, and recent developments in the field of UPQC, while [14] presents a survey on the widespread application of DG through various renewable energy systems (RES), PQ problems related to the power electronic interface, and the use of the UPQC to improve PQ and system reliability.

In contrast to these surveys, this paper provides an exhaustive review of topologies, control strategies and algorithms, optimal methods, and applications. In other words, this paper presents a combination of [14], [16], and [20] state-of-the-art publications. For this study, various PQ standards, books, book chapters, IEEE transactions, peer-reviewed journals,



Fig. 2: Published literature on the UPQC.

and conference proceedings have been examined to identify the various UPQC features and recent technological improvements. Fig. 1 presents details of the previous literature reviewed for this paper.

Fig. 2 shows the year-wise publication of the literature studied. As can be observed from Fig. 2, the UPQC is a popular topic among researchers due to its simultaneous series and shunt compensation features. In past years, researchers have attempted to find various applications and ways of improving UPQC performance.

This paper contributes to the existing literature by providing a detailed discussion on

1. Various UPQC topologies and compensation methods.

2. Time-domain control theories used for controlling the UPQC in various applications.

3. Various algorithms implemented for the optimal design of the UPQC based on kVA rating.

4. UPQC applications for the integration of RES.

5. Various algorithms implemented to determine the optimal location of the UPQC for minimum power loss in the distribution system.

This paper is organized as follows. The different UPQC topologies used in the literature are explained in Section 2. The various control strategies related to the UPQC are highlighted in Section 3. Section 4 covers the optimal designs based on the kVA rating. Section 5 presents the various applications for the UPQC in the integration of a solar PV system in RES, while the optimal location for the UPQC in the distribution system is explained in Section 6. Finally, Section 7 discusses the findings of this study while Section 8 presents the conclusion.

2. UPQC TOPOLOGIES

In the existing UPQC literature, various related topologies have been presented. A basic block diagram for the UPQC is shown in Fig. 3, consisting of a DVR in series, with the DSTATCOM in a shunt position to the AC power supply, shunt coupling inductor, DC-link, and a series injection transformer.

Topology	Description	Ref.
Right Shunt UPQC	• A DVR is connected in series with the AC mains and the DSTATCOM near to the	[3]
(UPQC-R)	load.	[0]
Left Shunt UPQC (UPQC-L)	• A DVR is connected on the load side and DSTATCOM near to the source.	[30,31]
Interline UPOC	• A DVR is linked with one of the feeders in series and DSTATCOM is connected in	[32]
(UPQC-I)	parallel with the second feeder.	
	• It can handle real power flow between two adjacent feeders.	
	• This arrangement only mitigates the current-related PQ problems effectively on the	
	feeder whereas, since the DSTATCOM is connected, the voltage related PQ problems	
	only occur on the feeder to which the DVR is connected.	
Multi-Converter	• An additional DC Source/VSC is connected to the DC- Link.	[33-35]
UPQC	• It shares real power between two adjacent feeders which are not connected.	
(UPQC-MC)	• It only mitigates the current-related PQ problems effectively on the feeder, whereas,	
	since DSTATCOM is connected, the voltage related PQ problems occur only on the	
	feeder to which the DVR is connected.	
Modular UPQC	• The VSC consists of numerous bridges connected in a cascade.	[36-40]
(UPQC-MD) &	• The DVRs of all cells are connected in parallel while DSTATCOMs are in series to	
Multilevel UPQC	form a UPQC-MD.	
(UPQC-ML)	• Multilevel converters are used for the UPQC-ML in a medium voltage distribution	
	system.	
Distributed UPQC	• The series transformer neutral of the DVR is utilized as a neutral for the 3P4W load.	[41]
(UPQC-D)	• The neutral current flowing toward the injecting transformer neutral is compensated	
	by adding the fourth leg of the DSTATCOM to the existing $3P3W$ UPQC while the	
	voltage of the transformer neutral point is retained at zero.	
Distributed	• The UPQC is integrated with the DG to the grid.	[42-48]
Generators UPQC	• The DG is connected to the DC-link of UPQC.	
(UPQC-DG)	• Power is supplied to the main grid via the DSTATCOM.	

Table 2: UPQC topologies based on configuration.



Fig. 3: UPQC block diagram.

2.1 UPQC Classification Based on Its Physical Structure

Various UPQC topologies have been proposed by researchers for different applications, and classified based on the converter, configuration of the DVR and DSTATCOM, and the supply system. All types of topologies are discussed in detail in the following subsections.

2.1.1 Converter Based Classification

The UPQC can be connected to the system using the voltage source converter (VSC) or current source converter (CSC). The VSC-based configuration is the most popular, due to its flexible control, high efficiency, and compact size [4]. The CSC-based UPQC is observed in [21–23]. The systematic non-linear control based on the state space variable model is used for controlling the CSC of the UPQC [21]. Kinhal *et al.* [22] considered the CSC-based UPQC in their performance analysis of the neural-network-based UPQC.

2.1.2 Configuration Based Classification

Depending on the connection of the DVR and DSTATCOM, the UPQC can be classified into various categories and used for specific applications. Significant works on various UPQC topologies based on the connection of the DVR and DSTATCOM are presented in Table 2. The right shunt UPQC (UPQC-R) is compact and simple to control. It requires a low rating converter and therefore has become more popular than the left shunt UPQC (UPQC-L) [24–25]. The UPQC utilized for the integration of RES into the main grid is known as UPQC-DG. This UPQC compensates for voltage-current-related PQ problems, maintaining synchronization between RES and the main grid as well as controlling the power generated by RES [14, 26–29]. This UPQC DC-link is fed by the power generated from RES.

According to Table 2, not only can the UPQC be

Topology	Highlighted Aspects	Ref.
Single-Phase	• DVR and DSTATCOM are isolated from each other by a bidirectional DC/DC-isolated converter	[63]
Two-Wire	and a high-frequency transformer.	
(1P2W)	• The power transfer can be controlled by adjusting the voltage phase shift between two inverters.	
Three-Phase	• In high voltage applications, the isolation transformer is used to isolate the DSTATCOM.	[64]
Three-Wire	• Ripple filters are connected across both the DVR and DSTATCOM to reduce the switching	[22], [29],
(3P3W)	ripples of the output.	[65–67]
	• In the case of 3P3W with an additional star-hexagon/T-connected transformer, neutral current	[68]
	mitigation is possible only if the voltages across these transformers are sinusoidal.	
Three-Phase	Split Capacitor (2C):	[69-72]
Four-Wire	• The neutral wire is connected at the mid-point of the capacitor which is at zero potential.	
(3P4W)	• The equal voltage across both capacitors avoids a circulating current.	
	• An additional control loop is added to maintain the DC-bus capacitor voltage regulation.	
	Four Leg (4L):	[73-76]
	• The neutral current load is compensated by an additional leg having two semiconductor switches.	
	Three H-Bridge (3-HB):	[24],
	• Three units of single-phase H-bridges are connected to the same DC-bus of the UPQC.	[77–79]
	• Without affecting its compensation capability, the UPQC works with reduced DC-link voltage as	[80]
	well as helping to fulfill the DC-link voltage requirement of the DSTATCOM and DVR.	

Table 3: Classification of the UPQC based on the supply system.

used to enhance PQ in the power distribution network but also to integrate the DG into the utility grid such as solar or wind power. Interconnected and islanding are the two operating modes of the UPQC-DG [13]. However, this topology has limitations such as interface issues, complex circuits, and high cost. The modular UPQC (UPQC-MD) and multilevel UPQC (UPQC-ML) are used in high power rating applications. Depending upon the operating voltage, the multilevel UPQC allows numerous H-bridge modules to be connected.

2.1.3 Supply System Based Classification

Based on the supply system, the UPQCs can be classified as single-phase two-wire (1P2W), threephase three-wire (3P3W), and three-phase four-wire (3P4W). Table 3 further illustrates the work carried out on types of UPQC based on the supply system. Various 1P2W and UPQC topologies are presented in [49], however, two half-bridges (HBs) with eightswitches are the most commonly used [26, 50–51]. In a three-leg (3L) topology with six switches, four switches are used for the DVR and two switches for DSTATCOM. However, in HB topology, two switches are used for the DVR and the remaining two switches for DSTATCOM [52].

In the case of a three-phase supply system, most of the research work has been conducted in a balanced environment using the 3P3W topology of the UPQC [53–60]. However, to mitigate 3P4W system problems such as excessive neutral current, harmonics, reactive power burden, and imbalance, the 3P4W UPQC tends to be implemented [61–62].

The performance of the UPQC depends upon the number of switching devices used for PEC. Therefore, 3L and HB topologies may affect the compensation performance of the UPQC. These topologies can be utilized for economically low power applications. The separate fourth leg used in four-leg (4L) topology controls the neutral current more effectively and therefore gives better performance than the split capacitor (2C) and 3-HB topologies for low to medium power applications. Since 12 switches are required in the 3-HB topology, it is more costly than the other two [81]. However, it is more suitable for medium to high power applications than 2C and 4L topology. In [82], 4L PEC is used for the DSTATCOM while 3L PEC is employed for the DVR with a split DC-link capacitor. In this study, the DC-bus voltage amplitude required for the DVR operation is less than that required for the DSTATCOM operation. Thus, it requires lower DC-bus voltage compared to the two 3L-2C topologies and a reduced number of switches in comparison to the two 4L topologies.

2.2 Classification Based on the Compensation Method

Based on the methods of compensation, the UPQC can be categorized as active power control (UPQC-P), reactive power control (UPQC-Q), simultaneous active and reactive power control (UPQC-S), and minimum volt-ampere (VA) loading (UPQC-VA_{min}). A detailed discussion of each method is presented as follows.

2.2.1 Active Power Control (UPQC-P)

In a UPQC-P, the DSTATCOM performs currentbased compensation with a unity PF at the load end, while the DVR performs voltage sag/swell compensation [83–87]. The operation of the UPQC-P during voltage sag in terms of its phasor diagram



Fig. 4: Phasor diagram of the UPQC-P for voltage sag.



Fig. 5: Phasor diagram of the UPQC-Q.

is represented in Fig. 4. It can clearly be observed that the UPQC-P requires only active power since the injected voltage is in phase with the supply current and supply voltage [36, 71].

The UPQC-P integrated with a wind turbine is used to provide active power to the grid as well as for mitigating PQ problems such as deep voltage sag and voltage interruption [25, 55]. The design and dynamic performance of the UPQC-P with solar PV to mitigate PQ problems and supply active power to the grid are explored in [46, 67].

2.2.2 Reactive Power Control (UPQC-Q)

The UPQC-Q requires only reactive power as the injected voltage since it is in quadrature with the supply current and supply voltage as presented in Fig. 5. Voltage sag compensation is achieved by the DVR without taking active power either from the source or PEC [54]. This DVR quadrature voltage injection leads the supply current which requires a large VAR loading of DVR [88–90]. Another limitation of the UPQC-Q is the phase difference between input and output voltage which relates to the severity of voltage sag and therefore it cannot effectively compensate for any kind of voltage drop



Fig. 6: Phasor diagram of the UPQC-P for voltage sag.

[66]. This can be overcome with minimum active power injection. For the compensation of higher magnitude voltage sags and voltage swell, a certain amount of active power is also required to be injected along with reactive power [66, 88]. A higher amount of series injection voltage is required by the UPQC-Q to compensate for the same amount of voltage sag in comparison to the UPQC-P [91].

2.2.3 Simultaneous Active and Reactive Power Control (UPQC-S)

In the UPQC-S, both active and reactive powers are delivered by the DVR and utilized to mitigate voltage sag/swell while load reactive power-sharing is achieved simultaneously by the DSTATCOM [59, 92– 93]. The phase angle control (PAC) approach can be used to control the UPQC-S for utilizing the UPQC to its maximum capacity [44, 45]. The operation of the UPQC-S during voltage sag and swell in terms of its phasors is shown in Fig. 6.

Under a supply voltage imbalance, effective operation of both PEC is still possible with equal reactive power-sharing. This is obtained by calculating the proper power angle (δ) [94]. The UPQC- S with solar PV shares the load reactive power requirement through the DVR during steady-state conditions as well as supplying active power to the grid, as illustrated in [95].

2.2.4 Minimum Volt-Ampere (VA) Loading (UPQC-VA_{min})

In similarity to the UPQC-S, both active and reactive powers are delivered by the DVR in UPQC-VA_{min}, and utilized to compensate for voltage sag/swell. However, this series voltage is added at a definite optimal angle by reference to the supply current [96–97]. The percentage of sag, PF angle, and power/swing angle (δ) decides the VA loading of the UPQC. The DVR is used to compensate for an imbalance in the voltage sag. Voltage injected at an optimal angle regulates it at the load end. Therefore, the DVR requires the minimum active power injection and hence the minimum VA loading of the UPQC [98]. Kolhatkar et al. [99] determined the optimum angle using a minimum VA optimization algorithm. The optimum phase angle and corresponding angles of the voltage injection for each case are calculated off-line from the 2D lookup table for percentage sag and PF angle. While in [72], an adaptive neuro-fuzzy inference system (ANFIS) is used to calculate the phase angle online. Kumar et al. [98] suggested the particle swarm optimization (PSO) technique for determining the optimum angle. The calculation of the optimum angle by considering the effects of source voltage THD and load current THD distortion is reported in [100]. An investigation into the fast-Fourier transform algorithm to find the optimum angle was carried out by Kisck et al. [97].

3. UPQC CONTROL STRATEGIES

Control strategies of UPQC include control theory and the controller used to control UPQC. UPQC mainly consists of two PECs connected in a backto-back fashion through a common DC-Link. The following subsections provide a detailed study of time-domain control theories used for controlling UPQC and various controllers used for DC-Link control.

3.1 Control Theories

In power electronics, control theory plays a very important role. The behavior, desired operation, and application of the UPQC for a specific system depend upon the control algorithm used. This, in turn, estimates the voltage signals and reference current, which determine the gating pulses of the VSC, to provide the preferred performance. Some of the control methods used by researchers are listed below:

- (a) frequency domain theories
- (b) time-domain theories
- (c) other techniques.

Frequency domain control theories based on the Fourier transform [101], Kalman filter [102], and Wavelet transform [103, 104] are not preferred for real-time control of compensators due to their sluggish and slow responses. They also involve substantial calculations [4]. On the other hand, time-domain theories are the most widely used for control purposes. Instantaneous active-reactive power or p-q and synchronous reference frame (SRF) or d-q theories are widely used in time-domain control algorithms. In these algorithms, voltages and current signals in the ABC-frame are transferred to the stationary reference frame (p-q theory) [105] or synchronously rotating frame (d-q) [64].

In p-q theory, instantaneous active and reactive powers are calculated from three-phase load currents and point of common coupling (PCC) voltages. The fundamental and fluctuating components of these powers are separated by using a filter. In most studies, a low pass filter (LPF) is used [74–79]. The referenced three-phase supply currents are estimated using the following expression.

$$\begin{bmatrix} i_{sa}\\ i_{sb}\\ i_{sc} \end{bmatrix} = \sqrt{\frac{2}{3}} \begin{bmatrix} 1 & 0\\ -\frac{1}{2} & \sqrt{\frac{3}{2}}\\ -\frac{1}{2} & -\sqrt{\frac{3}{2}} \end{bmatrix} \begin{bmatrix} v_{\alpha} & v_{\beta}\\ -v_{\beta} & v_{\alpha} \end{bmatrix} \begin{bmatrix} P\\ Q \end{bmatrix} \quad (1)$$

In SRF theory, Park's transformation and reverse Park's transformation are used for the conversion. These signals are synchronized with the PCC voltage using a phase-locked loop (PLL). The fundamental and harmonic components of d-q axis currents are separated using filters. This theory is mostly used for referencing signal generation. In [44], SRF-based control theory is used to control the UPQC integrated with solar PV while in [88] it is used to control the UPQC-Q with minimum active power injection. It is also used in the comparison of various 3P4W UPQC topologies [68].

The instantaneous power angle (δ) depends upon the variation in the source voltage. Another effective method of UPQC control is the PAC as presented in [92], whereby the instantaneous power angle (δ) is estimated from the extracted instantaneous load active and reactive power as given below

$$\delta = \sin^{-1} \frac{\text{Reactive Power Handled by DVR}}{\text{Load Active Power}} \quad (2)$$

However, this theory can only be implemented in a balanced system. Table 4 shows the time-domain control theories used in the literature.

According to the literature presented in Table 4, it can be clearly observed that under an unbalanced and distorted source voltage environment, performance of the p-q theory is unsatisfactory compared to SRF theory. The SRF method is superior to p-qtheory for the compensation of reactive power and current harmonics. In the PAC method, the DVR is controlled to share reactive power during voltage sag as well as during steady-state conditions. However, since the reactive power shared by the DVR varies with the voltage sag/swell, power balance can be maintained by controlling the appropriate power angle (δ).

3.2 UPQC Controller

The performance of the UPQC also depends upon the control of the DC-link. A DC-link feedback controller should respond quickly to re-establish the DC-link voltage to a fixed reference value with

Control Theory	Highlighted Aspects	Ref.
<i>p-q</i>	• Single-phase $p-q$ theory is implemented for load imbalance (3P4W system).	[41, 81,
Theory	• The referenced load voltage extracted for the DVR is utilized as a substitute for actual load	106]
	voltages to improve performance during distorted or imbalanced source voltages.	
	• Single-phase PLL is used to eliminate the effect of distorted supply voltage on the DSTATCOM	[107]
	performance.	
	• In single-phase p - q theory to maintain the DC-link voltage without using a PI-controller, fewer	[108]
	sensors are required.	[105 100]
	 In the modified <i>p-q</i> theory-based control algorithm, the supply current is controlled indirectly. Variable PAC is used based on the data driven control technique. 	[105, 109]
CDE	• variable FAC is used based on the data-driven control technique.	[110]
Shr	• The SKF-based control algorithm is used to control the OPQC in dual compensating mode.	[46, 111]
1 neory	• The DVR is controlled as a sinusoidal current source and the DSTATCOM as a voltage source.	[07]
	• The moving average filter (MAF) is used to extract the fundamental component of the <i>a</i> axis load current.	[67]
	• A synchronous double reference PLL frame is used to detect a positive sequence voltage in	[112]
	utility imbalance conditions.	
	• A PLL scheme is presented with less UPQC control using a zero-crossing detection-based line	[113]
	frequency synchronization technique.	
	• This saves the processor time and does not involve extra hardware.	
	• MAF is used to separate the positive sequence component.	
	• The MC-UPQC is controlled using the SRF algorithm, and the second-order LPF for extracting	[114]
	the fundamental component of the d axis load current.	
	• The fuzzy logic controller (FLC) is used to generate the d axis reference current of the	
	DSTATCOM.	
	• A PLL-less SRF control algorithm is used to control solar PV integrated UPQC.	[115]
	• A discrete filter is used for extraction of the fundamental positive sequence (FPS) components	
	of grid voltage.	
Phase Angle	• In the fixed PAC method, the power angle (δ) is maintained constant under all operating	[106]
Control	conditions, whereas in the variable PAC method, the power angle (δ) is kept variable to allow	
(PAC)	for voltage fluctuation.	
Method	• PAC and instantaneous symmetrical component theory-based approaches are used to control	[116]
	the UPQC.	
	• The power angle (δ) is estimated using the simple mathematical approach of "triangle rule of	
	vector addition".	
	• PAC-based SRF with a modified PLL algorithm is used for highly distorted and unbalanced	[76, 94,
	conditions.	117]
	• The power angle (δ) estimate is based on equal reactive power-sharing, with efficient utilization	
	of both inverters under a supply imbalance.	
	• A PAC for an unbalanced load is presented.	[118]
	\bullet Balanced reactive power shared by the DVR and unbalanced reactive power greater than 50%	
	of the total reactive power is shared by the DSTATCOM.	

Table 4: Time-domain control theories for the UPQC.

the least delay while exceeding it during dynamic conditions. The most widely used DC-link controller is the proportional integral (PI) based controller [59, 66, 83, 119]. The primary requirements of a UPQC controller are quick detection of the disruption, high precision, fast-tracking of the reference signal, and a high dynamic response. However, due to parameter variations and the non-linearity of load disturbance, the conventional controller fails to perform.

This problem is overcome by implementing a fuzzy logic-based PI-controller [17, 120–121], artificial neural network (ANN) based controller [122], the fuzzy- proportional integral derivative (PID) controller [123], and an ANFIS [124]. An ANN-based controller is fast in response and maintains the

stability of both converters for wide operating ranges [22]. The optimum value gained by the PI-controller improves system performance. In [125], the FLC along with the ant colony optimization technique is used. In [126], the PSO along with fuzzy based PI control are used to control the solar PV-UPQC. The virus colony search algorithm [127] and flower pollination optimization algorithm [128] are implemented to determine the optimum value of the PI-controller gains to mitigate voltage sag, THD, and improve the power of the hybrid system using the UPQC. A fuzzy-PID controller in the UPQC is used to control the DC-link capacitor voltage since it combines the benefits of the high steady-state accuracy of PID control and good robustness of the fuzzy control [123]. Both fuzzy and neural network systems are incorporated the ANFIS to estimate the referenced DC-link voltage and DC-link voltage regulation [124]. To minimize the UPQC online VA loading, a data-driven control (DDC) technique is developed using the variable PAC method [110]. In [93], the gravitational search algorithm is used to tune the PI-controller gains of the UPQC-S.

4. OPTIMUM UPQC DESIGN BASED ON KVA RATING

The DVR is designed to handle voltage-related PQ problems and otherwise remains ideal in steady-state conditions. Therefore, the DSTATCOM is loaded heavily, increasing the overall size of the UPQC and reducing the reliability of the system. The DSTATCOM plays an important role in maintaining the power system balance in all conditions. The kVA ratings of the DVR and DSTATCOM are dependent upon the load VAR variation and percentage of sag/swell. The kVA rating of the UPQC depends upon the compensation method. A steady-state analysis of the UPQC is given in [86]. The kVA rating issues are discussed in [24]. In [129], an attempt is made to reduce the DC-link voltage, resulting in an overall reduction in the VA rating of the PE converter and switches. To reduce the VA loading of DSTATCOM and the overall VA rating of the UPQC, many researchers have tried to use a DVR to achieve reactive power compensation. A DVR can be used not only to mitigate voltage sag/swell but also to compensate for a certain amount of load reactive power, thereby increasing DVR utilization and reducing the kVA rating of DSTATCOM [96]. The suitable angle at which voltage can be injected by the DVR is pre-calculated. This also reduces power loss in the UPQC [130]. The reactive power shared by the DVR depends upon the power angle (δ). Different techniques for estimating an optimum power angle (δ) have been reported in the literature. For instance, [76] discusses the PSO technique for minimizing the VA loading of the UPQC while considering the effect of source voltage and load current harmonics. In [43], the PAC algorithm is used to reduce the burden on the DSTATCOM by utilizing the DVR for active power injection into the system even during steady-state conditions, thereby helping to improve system reliability. In [131], the optimal size of the UPQC is determined using the following generalized equations. Total VA loading of the UPQC, as a function of the power angle (δ) and ratio between actual and rated source voltages (k), is represented by

$$S_{UPQC}\left(\delta,k\right) = S_{sh}\left(\delta,k\right) + S_{se}\left(\delta,k\right) \tag{3}$$

where VA loadings of the DVR and DSTATCOM in any operating conditions are computed by Eqs. (4) and (5), respectively

$$S_{se}\left(\delta,k\right) = \sqrt{\left[P_{se}\left(\delta,k\right)\right]^{2} + \left[Q_{se}\left(\delta,k\right)\right]^{2}} \qquad (4)$$

$$S_{sh}\left(\delta,k\right) = \sqrt{\left[P_{sh}\left(\delta,k\right)\right]^2 + \left[Q_{sh}\left(\delta,k\right)\right]^2} \quad (5)$$

A novel algorithm is presented to find the least possible VA rating of the UPQC as a function of the corresponding optimal power angle (δ) , DVR, DSTATCOM, and series transformer ratings. It is implemented for all operating conditions and UPQC compensation approaches. The interior-point method is used to determine the optimum size for UPQC converters to minimize the overall cost of the UPQC and series transformer [110, 132]. In [133], a two-stage optimization control algorithm based on the variable PAC method is proposed for the optimal design of UPQC. In a two-level algorithm, both the transformer VA rating and capital cost are considered for optimization [134]. In [135], a superconducting fault current limiter (SFCL) is connected to the parallel feeder of the main feeder where the UPQC is connected. The SFCL limits the excessive fault current, resulting in a considerable reduction in the UPQC rating. In [136], a normalized simulated annealing algorithm is compared with the analytical hierarchy process objective optimization to find the optimal solution. The optimization model contains the minimization of VA loading, SFCL conductance, and temperature within the constraints of the referenced voltage load.

5. APPLICATION OF THE UPQC IN RES

Distributed generation (DG) offers a gateway to enhance power system reliability while decreasing the requirement for forthcoming growth. The fast development of PEC technology makes it possible to integrate DG into the main grid through gridinteractive inverters; however, it deteriorates the PQ [137]. This is overcome by using the UPQC to integrate the RES with the main grid. Proper control of the DC-link voltage for the UPQC improves the dynamic performance of the system during disturbances. In [138], a hybrid RES consisting of fuel cell and doubly-fed induction generation-based wind energy systems are integrated with a weak grid using the UPQC. In [137], the UPQC is utilized for the integration of RES with the main grid, employing a PI-controller with FLC. To integrate a wind energy system with improved PQ, the Firefly algorithm is used to optimize the control pulses with a recurrent neural network for training the optimization parameter [139]. In this study, the focus is on applying the UPQC to integrate solar PV with the main grid. In a solar PV integrated UPQC, the array is directly connected to the DC-link of the solar PV-UPQC. The reference voltage for the DC-bus is obtained from the maximum power tracking (MPPT) algorithm. The DSTATCOM of the solar PV-UPQC performs dual functions such as compensating the reactive power while providing active power from the PV array into the grid. The DVR of the SPV-UPQC mitigates the sags and swells. In [140], an attempt is made to optimize the solar PV hosting capacity and energy loss in the distribution system by placing the open UPQC in the proper place. The solar PV-UPQC with a battery can supply power during peak periods. An analysis of the open solar PV-UPQC with and without a battery is presented in [141]. A biogeography-based optimization (BBO) technique for harmonics elimination in a smart grid integrated with solar PV using the UPQC is presented in [18]. The optimum switching angle of the converter eliminates the lower-order harmonics, while the higher-order harmonics are blocked by injecting an equal magnitude and opposite phase of the same order harmonics from the PE converter. In [46], the dynamic performance is presented of a solar PV integrated UPQC-P under varying irradiation, voltage sag, swell, and a step-change in load. The design and performance investigation of a 3P3W solar PV-UPQC is witnessed in [67], whereby the UPQC is designed using the following mathematical expressions.

Voltage magnitude of the DC-link:

$$V_{dc} = \frac{2\sqrt{2} \cdot V_{LL}}{\sqrt{3}m} \tag{6}$$

where V_{LL} is the line voltage of the grid and m is the depth of modulation.

DC-link capacitor rating:

$$C_{dc} = \frac{3 \cdot k \cdot a \cdot V_{ph} \cdot i_{DSTAT} \cdot t}{0.5 \left(V_{dc}^2 - V_{dc1}^2\right)} \tag{7}$$

where a is the overloading factor and t is the time required to attain a steady value after a disturbance.

Interfacing inductor for the DSTATCOM:

$$L_f = \frac{\sqrt{3} \cdot m \cdot V_{dc}}{12 \cdot a \cdot f_{sh} \cdot I_{cr,pp}} \tag{8}$$

Interfacing inductor of the DVR:

$$L_r = \frac{\sqrt{3} \cdot m \cdot V_{dc} \cdot K_{se}}{12 \cdot a \cdot f_{se} \cdot I_r} \tag{9}$$

where f_{se} and f_{sh} are the switching frequencies of the DVR and DSTATCOM, respectively, K_{se} is the transformation ratio of the series transformer.

In [142], a control approach is discussed, based on a second-order generalized integrator (SOGI) and delayed signal cancellation (DSC). Two cascaded SOGI (CSOGI) band-pass filters are used for the extraction of the PCC voltage template to control the DVR. The DSTATCOM control is based on the extraction of the fundamental load current using CSOGI. This method extracts the fundamental component of the load current with better accuracy and improved dynamic performance. A single-phase solar PV-UAPF with UPQC is proposed in [117] to improve the dynamics without affecting the steady-state performance. The performance of the UPQC-S incorporated with solar PV under distorted PCC voltages with a modified p-qtheory-based control algorithm is presented in [95]. A generalized cascaded delay signal cancellation technique is used for the extraction of the fundamental positive sequence components of voltages in the PCC. Integrated solar PV at the DC-link of PV-UPQC-S decreases the load demand on the main supply. In [143], a digital adaptive notch filter (DANF) is used to extract the fundamental active current and grid voltage template. The DANF is a good alternative to the PLL. In [115], a digital filter is used to extract the fundamental positive sequence component of the distorted grid voltage which is used to generate the sine and cosine values necessary for d-q based control.

6. UPQC LOCATION IN THE DISTRIBU-TION NETWORK

The performance of the distributed system in terms of improving voltage stability, voltage regulation, reducing power loss and overall system costs, along with maximizing system reliability, can be achieved by optimal reactive power compensation. The effectiveness of optimal reactive power compensation can be improved by placing the UPQC at the proper location [144].

Based on the size of the UPQC, the optimal location in the radial distribution system can be determined using the differential evolution algorithm [145]. In the three-phase unbalanced distribution network, the Cuckoo optimization algorithm is implemented to allocate the optimal place for the UPQC [146]. In [147], identifying the UPQC location in the radial distribution system is framed as a non-linear single-objective problem to decrease the real power loss and improve the voltage profile of the system. In this regard, the latest technique, i.e., the Ant Lion optimization algorithm is implemented. The multi-objective Whale optimization algorithm can be effectively implemented to determine the optimal location and size of DGs in distribution systems for minimum power loss [148]. The multi-objective Grey Wolf optimizer algorithm based on probabilistic load flow along with the fuzzy method for choosing the best final answer is implemented to find the optimal location and size for the PAC-based UPQC [149]. In [150], the Grasshopper optimization algorithm (GOA) is proposed to find the optimal location and size for the UPQC. The GOA is a novel optimization method for encouraging the movement and immigration of a grasshopper in nature. It contains the global convergence requirement, and also supports the capabilities of local and global searches. A meta-heuristic Firefly algorithm is proposed in [151] to determine the optimal size for the DG in

Optimization	Highlighted Aspects	Ref.
Algorithm		
PSO	• Higher loading ability	[155]
	• Can be implemented on non-linear loads	
	• Requires the consideration of various types of modeling	
	• Applicable for use in the science and engineering fields	[156]
	• Requires no mutation and overlapping calculation	
	• Can be implemented on a single-phase load	
Differential	• Reduced voltage fluctuation enhances speed	[157]
Evolution	• Can be implemented on baseloads	
	• Line loading limits need to be considered	
Imperialist	• Maintains energy supply continuity	[158]
Competitive	• Minimizes consumer discontent	
Algorithm	• Can be implemented on linear loads	
(ICA)	• Lower convergence rates	
	• Cannot solve the multi-objective problem	
Firefly Algorithm	• Cost-effective and easy to apply	[159]
Differential	• Can be implemented on non-linear loads	
Evolution	• Requires further observation on power loss	
Multi-Objective	• Enhances the PV hosting capacity	[140]
PSO	• Reduces energy loss	
(MOPSO)	• Can be implemented on linear loads	
	• Requires inspection on various converters	
Moth-Flame	• DVR voltage is determined using MFO	[160]
Optimization	• DVR is utilized for reactive power compensation	
Pareto-based	• Increases the peak load shaving	[161]
MOPSO	• Minimizes the total power loss	
	• Can be implemented on non-linear loads	
Demand	• Minimizes power loss	[162]
Response	• Can be used to identify optimal electricity prices in DRP during peak hours	
Programs	• Consideration is given to responsive loads	
	• Requires frequency control and congestion management	

 Table 5: Optimization techniques for identifying the appropriate location and size of the UPQC.

the power system. The loss-sensitive (LS) method is used to determine the appropriate locations of DG. Accordingly, the LS index is calculated such that the variation in power loss is divided with rises in the generation and rank of buses to recognize the appropriate positions for DG. Optimization models implemented to determine the location of the UPQC must offer more reliability and feedback signals. Some hybrid algorithms have been used to determine the optimal location of the UPQC by focusing on efficiency, system cost, and voltage stability index such as a hybrid algorithm of the genetic algorithm and dragonfly algorithm (DA) known as the genetically modified DA algorithm [152] and crow search mating-based lion algorithm [153]. In [154], a novel rider optimization algorithm (ROA)-modified PSO on a fitness basis is presented for the improvement of PQ by choosing a particular location and size for the UPQC. Accordingly, a steady-state model of UPQC is derived to set the forward/backward sweep load flow with the objective of enhancing voltage and current profiles, while minimizing power loss and the investment costs. Table 5 summarizes the various algorithms used in the literature to determine

the optimal location and size for the UPQC in the distribution system.

7. DISCUSSION

The recent research carried out in the field of UPQC to develop new topologies, compensation methods, control theories, and controllers is presented in Fig. 7. It can clearly be observed from Fig. 7 that since the introduction of UPQC for PQ enhancement, researchers have attempted to implement it in various applications with the objective of mitigating PQ problems and integrating RES with the main grid. However, the UPQC is still in the research phase and has not yet been extensively applied. The main reason for this is that the UPQC includes two sets of PE converters and a series transformer. This raises the overall cost dramatically along with production and marketing expenses, etc. The main challenge to the researcher is reducing the VA rating of the UPQC without compromising on performance.

Many studies have been conducted to identify the features of the UPQC for its application in a balanced environment. However, the power system is

	1998	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020
Topologies(Physical																						
Structure)																						
Topologies(Compensation																						
Method)																						
Control Theories																						
Controllers Used																						
Optimum Design Based on kVA Rating																						
Application in RES																						
Location of UPQC																						

Fig. 7: UPQC research.

unbalanced with both single-phase and three-phase loads. The major PQ problems occurring in an unbalanced power system are an excessive neutral current, harmonics, a reactive power burden, etc. According to the published literature in the last five years, it can be observed that researchers are attempting to implement a 3P4W UPQC to overcome these obstacles.

Researchers made a concerted effort to develop an effective and efficient technique to compensate for voltage sag/swell with the introduction of UPQC-P, UPQC-Q and UPQC-S topologies. However, to minimize costs, it is necessary to increase the utilization of both PECs during all operating conditions. According to the recent research articles, it is clear that compensation techniques and design algorithms are being developed for the maximum utilization of both PECs in all operating conditions. The UPQC performance mainly depends on the control algorithm employed. Focusing on the literature involving control theories, it can be determined that time-domain theories are more efficient compared to the frequency domain. The SRF and p-q theories appear to be more popular among researchers. However, the main drawback of SRF is the presence of the double harmonic component in the d axis current. Using LPF at a very low cut-off frequency helps to improve the performance at the cost of poor dynamics. This can be overcome by implementing MAF or digital filters.

The p-q theory has been widely used to control the UPQC-DG in the integration of RES. The main advantages of p-q theory-based control are that it is simple, fast, and does not require a complex computational block such as PLL. However, the performance of p-q theory is poor in unbalanced grid voltage conditions.

To extract the distorted components of voltage and current, various methods have been applied by researchers such as zero-crossing detection, adaptive notch filtering, least-squares estimation, complex vector filtering, and SOGI-PLL. Among them, the PLL block is widely used and more popular for its simple structure, effectiveness, and overall performance acceptability. However, the precision and rapidity of the PLL declines under extremely unstable and distorted circumstances. With the conventional PLL algorithm, an inaccurate transformation angle is derived which results in undesired reference signal generation. However, with a modified PLL algorithm, a distortion-free transformation angle can be derived, providing proper synchronization of the reference signals with the supply system. Some advanced PLL algorithms are available that perform well in distorted circumstances such as filtering-based approaches, decoupling network-based PLLs, modified loop filter PLLs, etc.

The PAC approach is helpful in the effective utilization of the UPQC since it introduces active and reactive power-sharing features into the distribution system. There are two approaches found in the literature based on PAC, the first use p-q theory and the second use SRF theory The sensing circuit required in the PAC-based SRF method is less, enabling the controller to operate smoothly under an unbalanced and highly distorted source voltage environment since it includes the estimation of reference load voltage. Whereas the PAC-based p-q controller requires the magnitude of series-injected voltage and individual phase of the system to be calculated if operated under the same system conditions. Consequently, the increased number of sensors makes the calculation more complex. The PAC-based SRF theory with an advanced PLL algorithm is considered superior in terms of addressing problems associated with the real-time management of voltage, current, and reactive power from the source as well as the load side under highly distorted conditions.

The DC-link voltage control plays a very important role in improving the performance of the UPQC. Therefore, a quick response to maintain the DC-link voltage constant with minimum delay in all operating conditions is the basic requirement of a DC-link controller. The PI-controller is the most widely used in the literature. However, it fails to perform well due to parameter variations and the non-linearity of load disturbance. This can be overcome by implementing advanced controllers such as the fuzzy logic-based PI, fuzzy-PID, ANFIS, and ANN-based. These controllers provide a fast dynamic response and maintain the stability of both PECs in a wide operating range. The tuning of PI-controller gains using any optimization algorithm reduces the complexity involved in determining the controller parameters and gives better performance compared to manual and conventional PI-tuning methods.

By increasing the utilization of DVR for activereactive power-sharing during steady-state, the overall VA rating of the UPQC can be minimized and system reliability improved. Steady-state analysis and the factors on which the VA rating of the UPQC is dependent are highlighted by many researchers. Active-reactive power shared by the DVR depends upon the power angle (δ) . It can be observed that the optimization of the VA loading of the UPQC has rarely been deliberated so far. There is no comprehensive approach for optimizing the rating of a UPQC. It is clear from the existing literature on the implementation of various methods for minimizing the VA loading of the UPQC that power angle (δ) is controlled directly or indirectly. Satisfactory results are obtained in a particular operating condition such as voltage sag. On the other hand, minimum VA ratings of the DVR, DSTATCOM, series transformer, and, thus, the overall UPQC system is independent of minimum VA loading at a certain operating condition. Variation in the individual VA loading of both PECs is not considered during all operating conditions.

From Fig. 7, it can be clearly observed that other than PQ enhancement, using the UPQC for the integration of RES with the main grid has become a more popular field of research in recent years. The UPQC is used to integrate various RES (such as weak wind farms, fuel cells, and solar PV) with the grid to provide clean, green energy for society. In this case, the UPQC can be operated in two modes, i.e., interconnected and islanding, to supply the critical load without interruption. Solar PV-UPQC is efficient in enhancing the PQ of the grid and maintaining the voltage regulation compared to conventional grid-connected PV systems. Although the existing literature reports on the design and operation of solar PV-UPQC, no serious focus has been given to UPQC performance enhancement. Moreover, interfacing issues, complex circuits, high costs, and fast dynamic response are the main challenges to researchers.

The proper location of the UPQC in the distribution network provides the optimal reactive power compensation, resulting in the improved performance of the distributed system in terms of increased efficiency, voltage stability, voltage regulation, a reduction in the VA rating, and overall system costs, with better system reliability. Although there are numerous optimization algorithms, including advanced techniques such as GOA, GWO, Sine Cosine, PSO, hybrid PSO, according to the literature, Ant Lion and multi-objective Whale optimization have been found to be more effective for determining the optimal location and size of DGs for minimum power loss.

According to a review of the recent literature, the scope of future UPQC research should include the following.

1. Application and performance improvement of the UPQC under power system imbalance to mitigate PQ problems such as excessive neutral current, harmonics, reactive power burden, etc.

2. Design and development of advanced PLL algorithms such as MAF-based PLL to improve UPQC performance under normal, abnormal, and harmonically distorted grid conditions.

3. Design and development of an advanced controller with optimization techniques such as the Jaya algorithm to improve the dynamic response and maintain the stability of both PECs over a wide operating range.

4. Determination of the optimum power angle (δ) for maximum UPQC utilization under any operating conditions using advanced optimization algorithms such as Teaching-learning-based optimization (TLBO), Jaya, and Rao.

5. Design and development of an adaptive controller to address all PQ problems and regulate DC-link voltages in the solar PV-UPQC.

6. Determination of the optimal location and size for integrated DGs using the UPQC with the main grid to achieve minimum power loss through advanced optimization techniques.

8. CONCLUSION

This paper presents a review of the UPQC-based topologies, their compensation, control theories, and optimal location in the distribution system. The need for DG continues to increase at a rapid rate, while fast-growing PEC technology makes it possible to integrate DG with a grid. The UPQC is capable of compensating for all types of supply voltage and load current PQ problems along with the integration of RES with the main grid. Comparing all the topologies, it can be concluded that the UPQC-DG, UPQC-MC, and UPQC-ML play very important roles in the power system for integrating RES with the grid. From the rigorous study of various control theories, it is clear that the performance of the UPQC depends on control methods, but complexity of the control approach presents a significant challenge. It can be observed that the PAC-based SRF approach of the UPQC helps to maximize DVR utilization by reactive power-sharing while mitigating sag/swell. This results in a rating reduction for the DSTATCOM without affecting the overall UPQC rating. This paper presents relevant information on the UPQC and may act as a potential data source for research scholars, manufacturers, and electrical utilities working in the field of power system research.

REFERENCES

- R. C. Dugan, M. F. McGranaghan, and H. W. Beaty, *Electrical Power Systems Quality*, NY, USA: McGraw-Hill, 1996.
- [2] N. G. Hingorani and L. Gyugyi, Understanding FACTS: Concepts and Technology of Flexible AC Transmission Systems, Piscataway, NJ, USA: IEEE Press, 2000.
- [3] A. Ghosh and G. Ledwich, *Power Quality Enhancement Using Custom Power Devices*, Norwell, MA, USA: Kluwer Academic, 2002.
- [4] B. Singh, A. Chandra, and K. Al-Haddad, *Power Quality: Problems and Mitigation Techniques*, Hoboken, NJ, USA: John Wiley & Sons, Inc., 2015.
- [5] Electromagnetic Compatibility (EMC) Part 2: Environment Section 1: Description of Environment - Electromagnetic Environment for Low Frequency Conducted Disturbances and Signalling in Public Power Supply Systems, IEC 61000-2-1, International Electrotechnical Commission, May 1990.
- [6] Electromagnetic Compatibility (EMC) Part 2: Environment Section 5: Classification of Electromagnetic Environments, IEC 61000-2-5, International Electrotechnical Commission, Sep. 1995.
- [7] IEEE Recommended Practices on Monitoring Electric Power Quality, IEEE Standard 1159-1995, Institute of Electrical and Electronics Engineers, Nov. 1995.
- [8] Guide on Methods of Measurement of Short Duration Transients on Low Voltage Power and Signal Lines, IEC 816, International Electrotechnical Commission, 1984.
- [9] IEEE Recommended Practices on Surge Voltages in Low-Voltage AC Power Circuits, IEEE Standard c62.41-1991, Institute of Electrical and Electronics Engineers, 1991.
- [10] Electromagnetic Compatibility (EMC) Part 4: Limits Section 7: General Guide on Harmonics and Inter-harmonics Measurements and Instrumentation for Supply Systems and Equipment Connected Thereto, IEC 61000-4-7, International Electrotechnical Commission, 1991.
- [11] IEEE Recommended Practices and Requirements for Harmonic Control in Electric Power Systems, IEEE Standard 519-1992, Institute of Electrical and Electronics Engineers, 1992.
- [12] Electromagnetic Compatibility (EMC) Part 4: Limits Section 15: Flicker Meter - Functional

and Design Specifications, IEC 61000-4-15, International Electrotechnical Commission, 1997.

- [13] S. K. Khadem, M. Basu and M. F. Conlon, "Intelligent Islanding and Seamless Reconnection Technique for Microgrid With UPQC," *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 3, no. 2, pp. 483–492, Jun. 2015.
- [14] N. D. Karelia and V. J. Pandya, "Distributed Generation and Role of UPQC – DG in Meeting Power Quality Criteria – A Review," *Proceedia Technology*, vol. 21, pp. 520–525, 2015.
- [15] M. T. L. Gayatri and A. M. Parimi, "Mitigation of supply & load side disturbances in an AC Microgrid using UPQC," in *IEEE 6th International Conference on Power Systems* (*ICPS*), 2016.
- [16] M. El-Habrouk, M. K. Darwish, and P. Mehta, "Active power filters: A review," *IEE Proceed*ings - Electric Power Applications, vol. 147, no. 5, pp. 403–413, Sep. 2000.
- [17] B. N. Singh, H. Chandra, K. Al-Haddad, and B. Singh, "Fuzzy control algorithm for universal active filter," in *Power Quality '98*, 1998, pp. 73–80.
- [18] S. Paramanik, K. Sarker, D. Chatterjee, and S. K. Goswami, "Smart Grid Power Quality Improvement Using Modified UPQC," in 2019 Devices for Integrated Circuit (DevIC), 2019, pp. 356–360.
- [19] N. Gowtham and S. Shankar, "UPQC: A Custom Power Device for Power Quality Improvement," *Materials Today: Proceedings*, vol. 5, no. 1, pp. 965–972, 2018.
- [20] V. Khadkikar, "Enhancing Electric Power Quality Using UPQC: A Comprehensive Overview," *IEEE Transactions on Power Electronics*, vol. 27, no. 5, pp. 2284–2297, May 2012.
- [21] P. E. Melín, J. R. Espinoza, J. A. Muñoz, C. R. Baier, and E. E. Espinosa, "Decoupled control of a Unified Power Quality Conditioner based on a current source topology for fast AC mains disturbance compensation," in 2010 IEEE International Conference on Industrial Technology, 2010, pp. 730–736.
- [22] V. G. Kinhal, P. Agarwal, and H. O. Gupta, "Performance Investigation of Neural-Network-Based Unified Power-Quality Conditioner," *IEEE Transactions on Power Delivery*, vol. 26, no. 1, pp. 431–437, Jan. 2011.
- [23] R. Dharmalingam, S. S. Dash, K. Senthilnathan, A. B. Mayilvaganan, and S. Chinnamuthu, "Power Quality Improvement by Unified Power Quality Conditioner Based on CSC Topology Using Synchronous Reference Frame Theory," *The Scientific World Journal*, vol. 2014, 2014, Art. no. 391975.
- [24] S. Chen and G. Joos, "Rating issues of

unified power quality conditioners for load bus voltage control in distribution systems," in 2001 IEEE Power Engineering Society Winter Meeting. Conference Proceedings, vol. 2, 2001, pp. 944–949.

- [25] H. Toodeji, S. H. Fathi, and G. B. Gharehpetian, "Power management and performance improvement in integrated system of variable speed wind turbine and UPQC," in 2009 International Conference on Clean Electrical Power, 2009, pp. 609–614.
- [26] J. M. Correa, S. Chakraborty, M. G. Simoes, and F. A. Farret, "A single phase high frequency AC microgrid with an unified power quality conditioner," in 38th IAS Annual Meeting on Conference Record of the Industry Applications Conference, vol. 2, 2003, pp. 956–962.
- [27] S.-W. Park, I.-Y. Chung, J.-H. Choi, S.-I. Moon, and J.-E. Kim, "Control schemes of the inverter-interfaced multi-functional dispersed generation," in 2003 IEEE Power Engineering Society General Meeting, vol. 3, 2003, pp. 1924– 1929.
- [28] B. Han, B. Bae, H. Kim, and S. Baek, "Combined operation of unified power-quality conditioner with distributed generation," *IEEE Transactions on Power Delivery*, vol. 21, no. 1, pp. 330–338, Jan. 2006.
- [29] A. Q. Ansari, B. Singh, and M. Hasan, "Power conditioning of Distributed Generation system using Unified Power Quality Conditioner," in 2015 Annual IEEE India Conference (INDI-CON), 2015.
- [30] N. G. Jayanti, M. Basu, M. F. Conlon, and K. Gaughan, "Performance comparison of a left shunt UPQC and a right shunt UPQC applied to enhance fault-ride-through capability of a fixed speed wind generator," in 2007 European Conference on Power Electronics and Applications, 2007.
- [31] N. Patnaik, A. K. Panda, and P. R. Mohanty, "Performance and comparative rating evaluation of single phase left shunt UPQC," in 2016 IEEE International Conference on Power Electronics, Drives and Energy Systems (PEDES), 2016.
- [32] A. K. Jindal, A. Ghosh, and A. Joshi, "Interline Unified Power Quality Conditioner," *IEEE Transactions on Power Delivery*, vol. 22, no. 1, pp. 364–372, Jan. 2007.
- [33] P. Li, Q. Bai, B. Zhao, and Y. Yang, "Power quality control center and its control method," in 2005 IEEE/PES Transmission & Distribution Conference & Exposition: Asia and Pacific, 2005.
- [34] P. Li, Q. Bai, and G. Li, "Coordinated control strategy for UPQC and its verification," in *IEEE Power Engineering Society General*

Meeting, 2006.

- [35] H. R. Mohammadi, A. Y. Varjani, and H. Mokhtari, "Multiconverter Unified Power-Quality Conditioning System: MC-UPQC," *IEEE Transactions on Power Delivery*, vol. 24, no. 3, pp. 1679–1686, Jul. 2009.
- [36] B. Han, B. Bae, S. Baek, and G. Jang, "New configuration of UPQC for medium-voltage application," *IEEE Transactions on Power Delivery*, vol. 21, no. 3, pp. 1438–1444, Jul. 2006.
- [37] I. A. Rubilar, J. R. Espinoza, J. A. Muñoz, and L. A. Morán, "DC link voltage unbalance control in three-phase UPQCs based on NPC topologies," in 2007 IEEE Industry Applications Annual Meeting, 2007, pp. 597–602.
- [38] J. A. Munoz, J. R. Espinoza, L. A. Moran, and C. R. Baier, "Design of a Modular UPQC Configuration Integrating a Components Economical Analysis," *IEEE Transactions on Power Delivery*, vol. 24, no. 4, pp. 1763–1772, Oct. 2009.
- [39] N. Farokhnia, S. H. Fathi, and H. R. Toodeji, "Voltage sag and unbalance mitigation in distribution systems using multi-level UPQC," in 2010 First Power Quality Conference, 2010.
- [40] S. A. González and M. I. Valla, "UPQC implemented with Cascade Asymmetric Multilevel Converters," *Electric Power Systems Research*, vol. 124, pp. 144–151, Jul. 2015.
- [41] V. Khadkikar and A. Chandra, "A Novel Structure for Three-Phase Four-Wire Distribution System Utilizing Unified Power Quality Conditioner (UPQC)," *IEEE Transactions on Industry Applications*, vol. 45, no. 5, pp. 1897– 1902, 2009.
- [42] R. A. Mastromauro, M. Liserre, and A. Dell'Aquila, "Single-phase grid-connected photovoltaic systems with power quality conditioner functionality," in 2007 European Conference on Power Electronics and Applications, 2007.
- [43] N. K. S. Naidu et al., "Coordinated active power control between shunt and series converters of UPQC for distributed generation applications," in 42nd Annual Conference of the IEEE Industrial Electronics Society (IECON 2016), 2016, pp. 3697–3702.
- [44] A. Patel, H. D. Mathur, and S. Bhanot, "A new and simple SRF based power angle control for UPQCdg to integrate solar PV into grid," in 2017 7th International Conference on Power Systems (ICPS), 2017, pp. 75–80.
- [45] A. Patel, H. D. Mathur, and S. Bhanot, "A new SRF-based power angle control method for UPQC-DG to integrate solar PV into grid," *International Transactions on Electrical Energy* Systems, vol. 29, no. 1, 2019, Art. no. e2667.

- [46] S. Devassy and B. Singh, "Dynamic performance of solar PV integrated UPQC-P for critical loads," in 2015 Annual IEEE India Conference (INDICON), 2015.
- [47] S. K. Khadem, M. Basu, and M. F. Conlon, "A comparative analysis of placement and control of UPQC in DG integrated grid connected network," *Sustainable Energy, Grids and Networks*, vol. 6, pp. 46–57, Jun. 2016.
- [48] L. B. G. Campanhol, S. A. O. da Silva, and A. O. Azauri, "A three-phase four-wire grid-connected photovoltaic system using a dual unified power quality conditioner," in 015 IEEE 13th Brazilian Power Electronics Conference and 1st Southern Power Electronics Conference (COBEP/SPEC), 2015.
- [49] A. Nasiri and A. Emadi, "Different topologies for single-phase unified power quality conditioners," in 38th IAS Annual Meeting on Conference Record of the Industry Applications Conference, vol. 2, 2003, pp. 976–981.
- [50] M. Basu, S. P. Das, and G. K. Dubey, "Experimental investigation of performance of a single phase UPQC for voltage sensitive and non-linear loads," in 4th IEEE International Conference on Power Electronics and Drive Systems (IEEE PEDS 2001) - Indonesia, vol. 1, 2001, pp. 218–222.
- [51] J. R. Reyes, J. R. Espinoza, and C. A. Sepúlveda, "Operating region of single-phase UPQCs," in 2005 IEEE 36th Power Electronics Specialists Conference, 2005, pp. 1726–1731.
- [52] J. Prieto, P. Salmerón, and R. S. Herrera, "A unified power quality conditioner for wide load range: Practical design and experimental results," in 2005 IEEE Russia Power Tech, 2005.
- [53] V. Khadkikar, A. Chandra, A. O. Barry, and T. D. Nguyen, "Conceptual study of unified power quality conditioner (UPQC)," in 2006 IEEE International Symposium on Industrial Electronics, 2006, pp. 1088–1093.
- [54] V. Khadkikar and A. Chandra, "A Novel Control Approach for Unified Power Quality Conditioner Q without Active Power Injection for Voltage Sag Compensation," in 2006 IEEE International Conference on Industrial Technology, 2006, pp. 779–784.
- [55] M. F. Farias, P. E. Battaiotto, and M. G. Cendoya, "Wind farm to weak-grid connection using UPQC custom power device," in 2010 IEEE International Conference on Industrial Technology, 2010, pp. 1745–1750.
- [56] Y. Pal, A. Swarup, and B. Singh, "Performance of UPQC for power quality improvement," in 2010 Joint International Conference on Power Electronics, Drives and Energy Systems & 2010 Power India, 2010.

- [57] K. Palanisamy, J. S. Mishra, I. J. Raglend, and D. P. Kothari, "Instantaneous power theory based unified power quality conditioner (UPQC)," in 2010 Joint International Conference on Power Electronics, Drives and Energy Systems & 2010 Power India, 2010.
- [58] A. Moghadasi, S. M. Torabi, and M. Salehifar, "Combined operation of the unified power quality conditioner with SFCL and SMES," in 2010 First Power Quality Conference, 2010.
- [59] V. Khadkikar and A. Chandra, "UPQC-S: A Novel Concept of Simultaneous Voltage Sag/Swell and Load Reactive Power Compensations Utilizing Series Inverter of UPQC," *IEEE Transactions on Power Electronics*, vol. 26, no. 9, pp. 2414–2425, Sep. 2011.
- [60] N. Hari, K. Vijayakumar, and S. S. Dash, "A versatile control scheme for UPQC for power quality improvement," in 2011 International Conference on Emerging Trends in Electrical and Computer Technology, 2011, pp. 453–458.
- [61] M. Kesler and E. Ozdemir, "Synchronous-Reference-Frame-Based Control Method for UPQC Under Unbalanced and Distorted Load Conditions," *IEEE Transactions on Industrial Electronics*, vol. 58, no. 9, pp. 3967–3975, Sep. 2011.
- [62] Y. Pal, A. Swarup, and B. Singh, "A control strategy based on UTT and I Cosθ theory of three-phase, fourwire UPQC for power quality improvement," *International Journal of Engineering, Science and Technology*, vol. 3, no. 1, pp. 30–40, 2011.
- [63] Y. Zhang, S. Zhang, and J. Chen, "The applications of bidirectional full bridge DC–DC isolated converter in UPQC," in 2008 International Conference on Electrical Machines and Systems, 2008, pp. 1916–1921.
- [64] A. Elnady and M. M. A. Salama, "New functionalities of an adaptive unified power quality conditioner," in 2001 Power Engineering Society Summer Meeting, 2001, pp. 295–300.
- [65] J. Liu, Y. He, Y. Ye, and X. Wang, "An unified scheme and respective algorithms for the control of DC-linked double converters in an universal power quality controller," *IEEE* 34th Annual Conference on Power Electronics Specialist, 2003. PESC '03., vol. 1, 2003, pp. 327–332.
- [66] W. C. Lee, D. M. Lee, and T. K. Lee, "New Control Scheme for a Unified Power-Quality Compensator-Q With Minimum Active Power Injection," *IEEE Transactions on Power Deliv*ery, vol. 25, no. 2, pp. 1068–1076, Apr. 2010.
- [67] S. Devassy and B. Singh, "Design and Performance Analysis of Three-Phase Solar PV Integrated UPQC," *IEEE Transactions on Industry Applications*, vol. 54, no. 1, pp. 73–81,

Jan.-Feb. 2018.

- [68] Y. Pal, A. Swarup, and B. Singh, "A comparative analysis of three-phase four-wire UPQC topologies," in 2010 Joint International Conference on Power Electronics, Drives and Energy Systems & 2010 Power India, 2010.
- [69] J. Prieto, P. Salmerón, J. R. Vázquez, and F. J. Alcántara, "A series-parallel configuration of active power filters for VAr and harmonic compensation," in 28th Annual Conference of the IEEE Industrial Electronics Society (IECON 2002), vol. 4, 2002, pp. 2945–2950.
- [70] M. I. M. Montero, E. R. Cadaval and F. B. Gonzalez, "Comparison of Control Strategies for Shunt Active Power Filters in Three-Phase Four-Wire Systems," *IEEE Transactions on Power Electronics*, vol. 22, no. 1, pp. 229–236, Jan. 2007.
- [71] T. Zhili and Z. Dongjiao, "Design of dc Voltage Controller for UPQC by Using its Small Signal Model," in 2010 International Conference on Electrical and Control Engineering, 2010, pp. 3572–3575.
- [72] G. S. Kumar, B. K. Kumar, and M. K. Mishra, "Mitigation of Voltage Sags With Phase Jumps by UPQC With PSO-Based ANFIS," *IEEE Transactions on Power Delivery*, vol. 26, no. 4, pp. 2761–2773, Oct. 2011.
- [73] G. Chen, Y. Chen, and K. M. Smedley, "Three-phase four-leg active power quality conditioner without references calculation," in Nineteenth Annual IEEE Applied Power Electronics Conference and Exposition, 2004. APEC '04., vol. 1, 2004, pp. 587–593.
- [74] C. A. Molina, J. R. Espinoza, E. E. Espinosa, F. A. Villarroel, and J. A. Hidalgo, "Refined control of an Unified Power Quality Conditioner under nonlinear and asymmetrical loads," in 2010 IEEE International Symposium on Industrial Electronics, 2010, pp. 2581–2586.
- [75] Y. Pal, A. Swarup, and B. Singh, "New control algorithms for three-phase four-wire unified power quality conditioner — a simulation study," *Electrical Power Quality and Utilisation Journal*, vol. XVI, no. 1, pp. 1–10, 2013.
- [76] N. Patnaik and A. K. Panda, "Performance analysis of a 3 phase 4 wire UPQC system based on PAC based SRF controller with real time digital simulation," *International Journal* of Electrical Power & Energy Systems, vol. 74, pp. 212–221, Jan. 2016.
- [77] A. Ghosh, A. K. Jindal, and A. Joshi, "A unified power quality conditioner for voltage regulation of critical load bus," 2004 IEEE Power Engineering Society General Meeting, vol. 1, 2004, pp. 471–476.
- [78] F. Ng, M. C. Wong, and Y. D. Han, "Analysis

and control of UPQC and its DC-link power by use of p-q-r instantaneous power theory," in *Proceedings. 2004 First International Conference on Power Electronics Systems and Applications*, 2004, pp. 43–53.

- [79] A. Moghadasi, S. M. Torabi, and M. Salehifar, "Performance evaluation of Unified Power Quality Conditioner with SMES," 2010 First Power Quality Conference, 2010, pp. 1–6.
- [80] S. B. Karanki, N. Geddada, M. K. Mishra, and B. K. Kumar, "A Modified Three-Phase Four-Wire UPQC Topology With Reduced DC-Link Voltage Rating," *IEEE Transactions* on *Industrial Electronics*, vol. 60, no. 9, pp. 3555–3566, Sep. 2013.
- [81] V. Khadkikar, A. Chandra, and B. Singh, "Digital signal processor implementation and performance evaluation of split capacitor, four-leg and three H-bridge-based three-phase four-wire shunt active filters," *IET Power Electronics*, vol. 4, no. 4, pp. 463–470, Apr. 2011.
- [82] R. A. Modesto, S. A. O. Silva, and A. A. Oliveira Jr, "Three-phase four-wire unified power quality conditioner structure with independent grid current control and reduced dc-bus voltage operating with inverted/dual compensating strategy," *International Transactions on Electrical Energy Systems*, vol. 30, no. 6, Jun. 2020, Art. no. e12380.
- [83] H.-J. Ryoo, G.-H. Rim, T.-J. Kim, and D. O. Kisck, "Digital-controlled single-phase unified power quality conditioner nonlinear and voltage sensitive load," in 30th Annual Conference of IEEE Industrial Electronics Society (IECON 2004), vol. 1, 2004, pp. 24–29.
- [84] V. Khadkikar, A. Chandra, A. O. Barry, and T. D. Nguyen, "Analysis of Power Flow in UPQC during Voltage Sag and Swell Conditions for Selection of Device Ratings," in 2006 Canadian Conference on Electrical and Computer Engineering, 2006, pp. 867–872.
- [85] B.-R. Lin and C.- L. Huang, "Implementation of a shunt-series compensator for nonlinear and voltage sensitive load," in 2006 CES/IEEE 5th International Power Electronics and Motion Control Conference, 2006.
- [86] R. Rajasree and S. Premalatha, "Unified Power Quality conditioner (UPQC) control using Feed Forward (FF)/ Feed Back (FB) controller," in 2011 International Conference on Computer, Communication and Electrical Technology (IC-CCET), 2011, pp. 364–369.
- [87] V. Khadkikar, A. Chandra, A. O. Barry, and T. D. Nguyen, "Power quality enhancement utilising single-phase unified power quality conditioner: digital signal processor-based experimental validation," *IET Power Electronics*, vol. 4, no. 3, pp. 323–331, 2011.

- [88] M.-S. Yun, W.-C. Lee, I. Suh, and D.-S. Hyun, "A new control scheme of unified power quality compensator-Q with minimum power injection," in 30th Annual Conference of IEEE Industrial Electronics Society (IECON 2004), vol. 1, 2004, pp. 51–56.
- [89] M. Basu, S. P. Das, and G. K. Dubey, "Performance study of UPQC-Q for load compensation and voltage sag mitigation," in 28th Annual Conference of the IEEE Industrial Electronics Society (IECON 2002), vol. 1, 2002, pp. 698–703.
- [90] M. Basu, S. P. Das, and G. K. Dubey, "Investigation on the performance of UPQC-Q for voltage sag mitigation and power quality improvement at a critical load point," IET Generation, Transmission & Distribution, vol. 2, no. 3, pp. 414–423, 2008.
- [91] Y. Pal and A. Swarup, "A Comparative Analysis of UPQC-P, UPQC-Q and UPQC-VA_{min} - a simulation study," in *Proceedings of the 2014 International Conference on Power Systems*, *Energy, Environment*, 2014, pp. 80–85.
- [92] V. Khadkikar and A. Chandra, "A New Control Philosophy for a Unified Power Quality Conditioner (UPQC) to Coordinate Load-Reactive Power Demand Between Shunt and Series Inverters," *IEEE Transactions on Power Delivery*, vol. 23, no. 4, pp. 2522–2534, Oct. 2008.
- [93] S. J. Alam and S. R. Arya, "Observer-based control for UPQC-S with optimized gains of PI controller," *International Transactions on Electrical Energy Systems*, vol. 30, no. 7, Jul. 2020, Art. no. e47233.
- [94] A. K. Panda and N. Patnaik, "Management of reactive power sharing & power quality improvement with SRF-PAC based UPQC under unbalanced source voltage condition," *International Journal of Electrical Power & Energy Systems*, vol. 84, pp. 182–194, 2017.
- [95] S. Devassy and B. Singh, "Modified pq-Theory-Based Control of Solar-PV-Integrated UPQC-S," *IEEE Transactions on Industry Applications*, vol. 53, no. 5, pp. 5031–5040, Sep.-Oct. 2017.
- [96] Y. Y. Kolhatkar, R. R. Errabelli, and S. P. Das, "A sliding mode controller based optimum UPQC with minimum VA loading," in 2005 IEEE Power Engineering Society General Meeting, vol. 1, 2005, pp. 871–875.
- [97] D. O. Kisck, V. Navrapescu, and M. Kisck, "Single-phase unified power quality conditioner with optimum voltage angle injection for minimum VA requirement," in 2007 IEEE International Symposium on Industrial Electronics, 2007, pp. 2443–2448.
- [98] G. S. Kumar, P. H. Vardhana, B. K. Kumar, and M. K. Mishra, "Minimization of VA

loading of Unified Power Quality Conditioner (UPQC)," in 009 International Conference on Power Engineering, Energy and Electrical Drives, 2009, pp. 552–557.

- [99] Y. Y. Kolhatkar and S. P. Das, "Experimental Investigation of a Single-Phase UPQC With Minimum VA Loading," *IEEE Transactions on Power Delivery*, vol. 22, no. 1, pp. 373–380, Jan. 2007.
- [100] G. S. Kumar, B. K. Kumar, and M. M. Kumar, "Optimal VA loading of UPQC during mitigation of unbalanced voltage sags with phase jumps in three-phase four-wire distribution system," in 2010 International Conference on Power System Technology, 2010.
- [101] A. Kazemi, A. Mokhtarpour, and M. T. Haque, "A new control strategy for Unified Power Quality Conditioner (UPQC) in distribution systems," in 2006 International Conference on Power System Technology, 2006.
- [102] K. H. Kwan, P. L. So, and Y. C. Chu, "An Output Regulation-Based Unified Power Quality Conditioner With Kalman Filters," *IEEE Transactions on Industrial Electronics*, vol. 59, no. 11, pp. 4248–4262, Nov. 2012.
- [103] M. Forghani and S. Afsharnia, "Online Wavelet Transform-Based Control Strategy for UPQC Control System," *IEEE Transactions on Power Delivery*, vol. 22, no. 1, pp. 481–491, Jan. 2007.
- [104] N. Karthikeyan, M. K. Mishra, and B. Kalyan Kumar, "Complex wavelet based control strategy for UPQC," in 2010 Joint International Conference on Power Electronics, Drives and Energy Systems & 2010 Power India, 2010, pp. 1–6.
- [105] T. Zhili, L. Xun, C. Jian, and K. Yong, "A New Control Strategy of UPQC in Three-phase Four-wire System," in 2007 IEEE Power Electronics Specialists Conference, 2007, pp. 1060– 1065.
- [106] V. Khadkikar, "Fixed and variable power angle control methods for unified power quality conditioner: operation, control and impact assessment on shunt and series inverter kVA loadings," *IET Power Electronics*, vol. 6, no. 7, pp. 1299–1307, Aug. 2013.
- [107] V. Khadkikar, B. N. Singh, and A. Chandra, "Generalised single-phase p-q theory for active power filtering: simulation and DSP-based experimental investigation," *IET Power Electronics*, vol. 2, no. 1, pp. 67–78, Jan. 2009.
- [108] Y. Pal, A. Swarup, and B. Singh, "A comparison of single-phase p-q theory and UVT based control algorithms for single-phase UPQC," in 2010 Annual IEEE India Conference (INDI-CON), 2010.
- [109] F.-Z. Wu and S.-P. Pei, "Analysis and Research of Control Strategy and Signal Detection Based

on UPQC," in 2010 2nd International Workshop on Intelligent Systems and Applications, 2010.

- [110] J. Ye, H. B. Gooi, and F. Wu, "Optimization of the Size of UPQC System Based on Data-Driven Control Design," *IEEE Transactions on Smart Grid*, vol. 9, no. 4, pp. 2999–3008, Jul. 2018.
- [111] R. A. Modesto, S. A. O. da Silva, A. A. de Oliveira, and V. D. Bacon, "A Versatile Unified Power Quality Conditioner Applied to Three-Phase Four-Wire Distribution Systems Using a Dual Control Strategy," *IEEE Transactions on Power Electronics*, vol. 31, no. 8, pp. 5503–5514, Aug. 2016.
- [112] P. Rodriguez, L. Sainz, and J. Bergas, "Synchronous double reference frame PLL applied to a unified power quality conditioner," in 10th International Conference on Harmonics and Quality of Power, 2002, pp. 614–619.
- [113] I. Axente, J. N. Ganesh, M. Basu, M. F. Conlon, and K. Gaughan, "A 12-kVA DSP-Controlled Laboratory Prototype UPQC Capable of Mitigating Unbalance in Source Voltage and Load Current," *IEEE Transactions* on Power Electronics, vol. 25, no. 6, pp. 1471– 1479, Jun. 2010.
- [114] C. B. Paduchuri, S. S. Dash, C. Subramani, and S. H. Kiran, "An Efficient Control Strategy Based Multi Converter UPQC using with Fuzzy Logic Controller for Power Quality Problems," *Journal of Electrical Engineering and Technol*ogy, vol. 10, no. 1, pp. 379–387, Jan. 2015.
- [115] S. Devassy and B. Singh, "PLL-less d-q control of solar PV integrated UPQC," in 2016 IEEE International Conference on Power Electronics, Drives and Energy Systems (PEDES), 2016.
- [116] A. Q. Ansari, B. Singh, and M. Hasan, "Algorithm for power angle control to improve power quality in distribution system using unified power quality conditioner," *IET Generation*, *Transmission & Distribution*, vol. 9, no. 12, pp. 1439–1447, Sep. 2015.
- [117] A. K. Panda, N. Patnaik and R. Patel, "Power quality enhancement with PAC-SRF based single phase UPQC under non-ideal source voltage," in 2015 Annual IEEE India Conference (INDICON), 2015.
- [118] A. Patel, H. D. Mathur, and S. Bhanot, "An improved control method for unified power quality conditioner with unbalanced load," *International Journal of Electrical Power & Energy Systems*, vol. 100, pp. 129–138, Sep. 2018.
- [119] Q.-N. Trinh and H.-H. Lee, "Improvement of unified power quality conditioner performance with enhanced resonant control strategy," *IET Generation, Transmission & Distribution*,

vol. 8, no. 12, pp. 2114–2123, Dec. 2014.

- [120] S. Chakraborty and M. G. Simões, "Fuzzy ARTMAP based forecast of renewable generation for a high frequency AC microgrid," in 31st Annual Conference of IEEE Industrial Electronics Society (IECON 2005), 2005, pp. 762–767.
- [121] M. Fatiha, M. Mohamed, and A.-A. Nadia, "New hysteresis control band of an unified power quality conditioner," *Electric Power Systems Research*, vol. 81, no. 9, pp. 1743–1753, 2011.
- [122] A. J. Laxmi, G. T. R. Das, K. U. Rao, K. Sreekanthi, and K. Rayudu, "Different control strategies for Unified Power Quality Conditioner at load side," in 2006 1st IEEE Conference on Industrial Electronics and Applications, 2006.
- [123] R. Liu, N. Xia, and X. Wang, "The research on fuzzy-PID control in unified power quality conditioner," in 2009 4th IEEE Conference on Industrial Electronics and Applications, 2009, pp. 821–824.
- [124] A. S. Kumar, S. Rajasekar, and P. A.-D.-V. Raj, "Power Quality Profile Enhancement of Utility Connected Microgrid System Using ANFIS-UPQC," *Procedia Technology*, vol. 21, pp. 112–119, 2015.
- [125] N. Kumarasabapathy and P. S. Manoharan, "MATLAB Simulation of UPQC for Power Quality Mitigation Using an Ant Colony Based Fuzzy Control Technique," *The Scientific World Journal*, vol. 2015, pp. 1–9, 2015.
- [126] S. Arulkumar and P. Madhavasarma, "Particle Swarm Optimization Technique Tuned Fuzzy Based PI Controller of Unified Power Quality Conditioner," *Journal of Computational* and Theoretical Nanoscience, vol. 14, no. 7, pp. 3433–3441, Jul. 2017.
- [127] B. Berbaoui, "Fuzzy multi-objective technique combined with VCS algorithm for unified power quality conditioner based on hybrid power source PEMFC/SC," *International Journal of Hydrogen Energy*, vol. 43, no. 12, pp. 6275– 6293, Mar. 2018.
- [128] K. Hussain and P. Sridhar, "Multi-Objective Flower Pollination Algorithm Based Controller for UPQC Including Hybrid Power Source," *International Journal of Innovative Technol*ogy and Exploring Engineering, vol. 9 no. 3, pp. 3493–3502, Jan. 2020.
- [129] V. Muneer, J. Sukumaran, and A. Bhattacharya, "Investigation on reduced DC link voltage based UPQC for harmonic compensation under unbalanced load," in 2017 International Conference on Technological Advancements in Power and Energy (TAP Energy), 2017.

- [130] M. Basu, M. Farrell, M. F. Conlon, K. Gaughan, and E. Coyle, "Optimal control strategy of UPQC for minimum operational losses," in 39th International Universities Power Engineering Conference, vol. 1, 2004, pp. 246–250.
- [131] B. B. Ambati and V. Khadkikar, "Optimal Sizing of UPQC Considering VA Loading and Maximum Utilization of Power-Electronic Converters," *EEE Transactions on Power Delivery*, vol. 29, no. 3, pp. 1490–1498, Jun. 2014.
- [132] A. Patel, S. K. Yadav, H. D. Mathur, S. Bhanot, and R. C. Bansal, "Optimum sizing of PV based UPQC-DG with improved power angle control," *Electric Power Systems Research*, vol. 182, May 2020, Art. no. 106259.
- [133] J. Ye, H. B. Gooi, and F. Wu, "Optimal Design and Control Implementation of UPQC Based on Variable Phase Angle Control Method," *IEEE Transactions on Industrial Informatics*, vol. 14, no. 7, pp. 3109–3123, Jul. 2018.
- [134] J. Ye, H. B. Gooi, X. Zhang, B. Wang, and U. Manandhar, "Two-Level Algorithm for UPQC Considering Power Electronic Converters and Transformers," in 2019 IEEE Applied Power Electronics Conference and Exposition (APEC), 2019, pp. 3461–3467.
- [135] A. H. Moghadasi, H. Heydari and M. Salehifar, "Reduction in VA rating of the Unified Power Quality Conditioner with superconducting fault current limiters," in 2010 1st Power Electronic & Drive Systems & Technologies Conference (PEDSTC), 2010, pp 382–387.
- [136] H. Heydari and A. H. Moghadasi, "Optimization Scheme in Combinatorial UPQC and SFCL Using Normalized Simulated Annealing," *IEEE Transactions on Power Delivery*, vol. 26, no. 3, pp. 1489–1498, Jul. 2011.
- [137] U. S. Vadivu and B. K. Keshavan, "Power quality enhancement of UPQC connected WECS using FFA with RNN," in 2017 IEEE International Conference on Environment and Electrical Engineering and 2017 IEEE Industrial and Commercial Power Systems Europe (EEEIC / I&CPS Europe), 2017.
- [138] G. Mallesham and C. S. Kumar, "Enhancement of power quality using UPQC for hybrid PEMFC and DFIG based wind energy system connected to weak grid," in 2017 International Conference on Technological Advancements in Power and Energy (TAP Energy), 2017.
- [139] T. Vigneysh and N. Kumarappan, "Grid Interconnection of Renewable Energy Sources Using Unified Power Quality Conditioner: A Fuzzy Logic-Based Approach," *Journal of Circuits, Systems and Computers*, vol. 28, no. 8, Jul. 2019, Art. no. 1950135.
- [140] S. Lakshmi and S. Ganguly, "Simultaneous optimisation of photovoltaic hosting capacity

and energy loss of radial distribution networks with open unified power quality conditioner allocation," *IET Renewable Power Generation*, vol. 12, no. 12, pp. 1382–1389, Sep. 2018.

- [141] S. Lakshmi and S. Ganguly, "Modelling and allocation of open-UPQC-integrated PV generation system to improve the energy efficiency and power quality of radial distribution networks," *IET Renewable Power Generation*, vol. 12, no. 5, pp. 605–613, Apr. 2018.
- [142] S. Devassy and B. Singh, "Control of solar photovoltaic integrated UPQC operating in polluted utility conditions," *IET Power Electronics*, vol. 10, no. 12, pp. 1413–1421, Oct. 2017.
- [143] S. Devassy and B. Singh, "Discrete Adaptive Notch Filter based Single Phase Solar PV Integrated UPQC," in 2016 IEEE 1st International Conference on Power Electronics, Intelligent Control and Energy Systems (ICPEICES), 2016.
- [144] K. Gholami, S. Karimi, and E. Dehnavi, "Optimal unified power quality conditioner placement and sizing in distribution systems considering network reconfiguration," *International Journal of Numerical Modelling: Electronic Networks, Devices and Fields*, vol. 32, no. 1, Jul. 2018, Art. no. e2467.
- [145] S. A. Taher and S. A. Afsari, "Optimal Location and Sizing of UPQC in Distribution Networks Using Differential Evolution Algorithm," *Mathematical Problems in Engineering*, vol. 2012, pp. 1–20, 2012.
- [146] J. Sarker and S. K. Goswami, "Optimal Location of Unified Power Quality Conditioner in Distribution System for Power Quality Improvement," *International Journal of Electrical Power & Energy Systems*, vol. 83, pp. 309–324, Dec. 2016.
- [147] M. L. Ramanaiah and M. D. Reddy, "Optimal Placement of Unified Power Quality Conditioner using Ant Lion Optimization Method," *International Journal of Applied Engineering Research*, vol. 12, no. 13, pp. 3708–3713, 2017.
- [148] A. Marimuthu, K. Gnanambal, and R. Priyanka, "Optimal allocation and sizing of DG in a radial distribution system using whale optimization algorithm," in 2017 International Conference on Innovations in Green Energy and Healthcare Technologies (IGEHT), 2017.
- [149] A. R. Moradi, Y. Alinejad-Beromi, M. Parsa, and M. Mohammadi, "Optimal Locating and Sizing of Unified Power Quality Conditioner-Phase Angle Control for Reactive Power Compensation in Radial Distribution Network with Wind Generation," *International Journal* of Engineering, Transactions B: Applications, vol. 31, no. 2, pp. 299–306, Feb. 2018.

- [150] M. Ebeed, S. Kamel, S. H. E. Abdel Aleem, and A. Y. Abdelaziz, "Optimal allocation of compensators," in *Electric Distribution Network Planning*, F. Shahnia, A. Arefi, and G. Ledwich, Eds., Singapore: Springer, 2018, pp. 321–353.
- [151] Z. B. Abdul Hamid, S. Jipinus, I. Musirin, M. M. Othman, and R. H. Salimin, "Optimal sizing of distributed generation using firefly algorithm and loss sensitivity for voltage stability improvement," *Indonesian Journal of Electrical Engineering and Computer Science*, vol. 17, no. 2, pp. 720–727, Feb. 2020.
- [152] K. Gaddala and P. S. Raju, "Optimal UPQC location in power distribution network via merging genetic and dragonfly algorithm," *Evolutionary Intelligence*, 2020.
- [153] K. Gaddala and P. S. Raju, "Merging Lion with Crow Search Algorithm for Optimal Location and Sizing of UPQC in Distribution Network," *Journal of Control, Automation and Electrical* Systems, vol. 31, no. 2, pp. 377–392, Feb. 2020.
- [154] K. Gaddala and P. S. Raju, "Optimal location of UPQC for power quality improvement: novel hybrid approach," *Journal of Engineering*, *Design and Technology*, vol. 18, no. 6, pp. 1519– 15541, 2020.
- [155] S. Ganguly, "Unified power quality conditioner allocation for reactive power compensation of radial distribution networks," *IET Generation*, *Transmission & Distribution*, vol. 8, no. 8, pp. 1418–1429, Aug. 2014.
- [156] S. Ganguly, "Multi-Objective Planning for Reactive Power Compensation of Radial Distribution Networks With Unified Power Quality Conditioner Allocation Using Particle Swarm Optimization," *IEEE Transactions on Power Systems*, vol. 29, no. 4, pp. 1801–1810, Jul. 2014.
- [157] S. Dutta, P. K. Roy, and D. Nandi, "Optimal location of UPFC controller in transmission network using hybrid chemical reaction optimization algorithm," *International Journal of Electrical Power & Energy Systems*, vol. 64, pp. 194–211, Jan. 2015.
- [158] M. Moazzami, M. J. Morshed, and A. Fekih, "A new optimal unified power flow controller placement and load shedding coordination approach using the Hybrid Imperialist Competitive Algorithm-Pattern Search method for voltage collapse prevention in power system," *International Journal of Electrical Power & Energy Systems*, vol. 79, pp. 263–274, Jul. 2016.
- [159] B. V. Kumar and N. V. Srikanth, "A hybrid approach for optimal location and capacity of UPFC to improve the dynamic stability of the power system," *Applied Soft Computing*, vol. 52, pp. 974–986, Mar. 2017.

- [160] M. L. Ramanaiah and M. D. Reddy, "Moth Flame Optimization Method for Unified Power Quality Conditioner Allocation," *International Journal of Electrical and Computer Engineering (IJECE)*, vol. 8, no. 1, pp. 530–537, Feb. 2018.
- [161] S. Lakshmi and S. Ganguly, "Multi-objective planning for the allocation of PV-BESS integrated open UPQC for peak load shaving of radial distribution networks," *Journal of Energy Storage*, vol. 22, pp. 208–218, Apr. 2019.
- [162] E. Dehnavi, S. Afsharnia, and K. Gholami, "Optimal allocation of unified power quality conditioner in the smart distribution grids," *Electrical Engineering*, vol. 101, no. 4, pp. 1277–1293, Nov. 2019.



Swati Gade received her B.E. and M.E. degrees in Electrical Engineering from the Pune University, India, in 2003 and 2007 respectively. Currently, she is an assistant professor in the Department of Electrical Engineering at Sandip Institute of Engineering and Management, Nashik, India. Also, she is Research Scholar at Sandip University Nashik, India. She has published about 6 national and international journal and

conference papers. Her research interest covers power systems, application of optimization techniques in power system, and power quality. She is a life member of ISTE, IE(i), and IAENGE (H.K.).



Rahul Agrawal received his B.E. Degree in Electrical Engineering from JIT Khargone, India in 2001, and M.E. in Power Electronics from SGSITS Indore, India in 2008, and Ph.D. from Maulana Azad National Institute of Technology Bhopal, India in 2017. Presently, he is an Associate Professor with the Department of Electrical and Electronics Engineering, Sandip University, Nashik, Maharashtra, India. He has 19 years

of teaching experience. He has published 20 papers in national and international conferences & journals. His current research interests are in the areas of application of optimization technique in power system, power electronics, and FACTS devices. He is a life member of ISTE, India.



Ravindra Munje received his B.E. degree in Electrical from Mumbai University, India in 2005, M.E. degree in Electrical (control systems) from Pune University, India in 2009, and Ph.D. Degree in Electrical from Swami Ramanand Teerth Marathwada University, Nanded, India in 2015. He was a Post-Doctoral Fellow with Shanghai Jiao Tong University, Shanghai, China from June 2017 to May 2019.

Currently, he is a Professor at the Electrical Engineering Department, K. K. Wagh Institute of Engineering Education and Research, Nashik, India. He has written a book and about 50 refereed papers. His research interests include modeling and the control of large-scale systems using sliding mode, multirate output feedback, power quality analysis using AI techniques.

Dr. Munje received the Promising Engineer Award in 2016, Outstanding Researcher Award in 2018 and Best Faculty Award in 2019.